

National College of Ireland

Project Submission Sheet

Student Name: Nandhini Viswarajan

Student ID: X23155540

Programme: MSc Entrepreneurship **Year:** 2024

Module: Dissertation

Lecturer: Venu Puthineedi (Supervisor)

Submission Due

Date: 10.08.2024

Project Title: Influence Of Government Policy on Fostering Entrepreneurial Mindset

at The Time of Economic Crisis: Comparing Ireland and India

Word Count: 14,278 Words

I hereby certify that the information contained in this (my submission) is information pertaining to research I conducted for this project. All information other than my own contribution will be fully referenced and listed in the relevant bibliography section at the rear of the project.

<u>ALL</u> internet material must be referenced in the references section. Students are encouraged to use the Harvard Referencing Standard supplied by the Library. To use other author's written or electronic work is illegal (plagiarism) and may result in disciplinary action. Students may be required to undergo a viva (oral examination) if there is suspicion about the validity of their submitted work.

Signature:

Date: 08.08.2024

PLEASE READ THE FOLLOWING INSTRUCTIONS:

- 1. Please attach a completed copy of this sheet to each project (including multiple copies).
- 2. Projects should be submitted to your Programme Coordinator.
- You must ensure that you retain a HARD COPY of ALL projects, both for your own reference and in case a project is lost or mislaid. It is not sufficient to keep a copy on computer. Please do not bind projects or place in covers unless specifically requested.
- 4. You must ensure that all projects are submitted to your Programme Coordinator on or before the required submission date. **Late submissions will incur penalties.**
- 5. All projects must be submitted and passed in order to successfully complete the year. Any project/assignment not submitted will be marked as a fail.

Office Use Only	
Signature:	
Date:	
Penalty Applied (if applicable):	

Al Acknowledgement Supplement

Dissertation

Influence Of Government Policy on Fostering Entrepreneurial Mindset at The Time of Economic Crisis: Comparing Ireland and India

Your Name/	Course	Date
Student Number		
Nandhini Viswarajan	Dissertation	08.08.2024
X23155540		

This section is a supplement to the main assignment, to be used if AI was used in any capacity in the creation of your assignment; if you have queries about how to do this, please contact your lecturer. For an example of how to fill these sections out, please click here.

AI Acknowledgment

This section acknowledges the AI tools that were utilized in the process of completing this assignment.

Tool Name	Brief Description	Link to tool
ChatGPT	ChatGPT is a natural language processing tool driven by AI technology that allows you to have human-like conversations and mucl more with the chatbot	

Description of Al Usage

This section provides a more detailed description of how the AI tools were used in the assignment. It includes information about the prompts given to the AI tool, the responses received, and how these responses were utilized or modified in the assignment. One table should be used for each tool used.

ChatGPT			
To check grammar, clarity, improvise the language and structure of the paragraph			

Evidence of Al Usage

This section includes evidence of significant prompts and responses used or generated through the AI tool. It should provide a clear understanding of the extent to which the AI tool was used in the assignment. Evidence may be attached via screenshots or text.

INFLUENCE OF GOVERNMENT POLICY ON FOSTERING ENTREPRENEURIAL MINDSET AT THE TIME OF ECONOMIC CRISIS: COMPARING IRELAND AND INDIA.

A dissertation submitted in partial fulfilment for the award of Master of Science in Entrepreneurship

By Nandhini Viswarajan X23155540

Supervisor Venu Puthineedi

National College of Ireland August 2024

ACKNOWLEDGEMENT

I owe a special debt of gratitude to my supervisor, Venu Puthineedi, for his invaluable guidance and mentorship. His insights and encouragement not only enhanced the quality of my thesis but also inspired me to publish my work.

I would also like to extend my heartfelt thanks to Keith Brittle for his unwavering support throughout the academic term. His assistance has been a cornerstone of my success.

My sincere thanks go to William Hogan and David Hurley for their timely help with data collection, which was crucial for my research.

Lastly, I am immensely grateful to all the participants who generously shared their valuable insights, making this study possible.

Nandhini Viswarajan August 2024

Table of Contents

ABSTF	RACT	3
1. IN	ITRODUCTION	4
2. LI	TERATURE REVIEW	7
2.1.	Impact of the Covid-19 Pandemic on Entrepreneurship	7
2.2.	Entrepreneurship and the Economy	10
2.3.	Role of Government in Promoting Entrepreneurship	13
2.4.	Entrepreneurship and Financial Independence	15
2.5.	Entrepreneurship in Developing and Developed Nations	18
3. RI	ESEARCH QUESTION	21
4. M	ETHODOLOGY	22
4.1.	Philosophy	22
4.2.	Research Design	24
4.3.	Participants and Sampling Size	27
4.4.	Data Collection Method	28
4.5.	Data Analysis	29
4.6.	Ethical Considerations	31
5. FI	NDINGS & ANALYSIS	32
5.1.	Difficulties in accessing the benefits of policy initiatives	32
5.2.	Financial support is essential to survive the business at the time of crisis:	34
5.3.	Entrepreneurial attitude is equally important to overcome the crisis	35
5.4.	Limitations of the Findings:	37
6. D	ISCUSSION	38
6.1.	Challenges in Accessing Government Benefits	38
6.2.	The Essential Role of Financial Support	39
6.3.	Sector-Specific Variations in Policy Effectiveness	39
6.4.	Comparative Analysis of Policy Approaches	40
6.5.	Implications for Future Policy Formulation	40
7. C	ONCLUSION & RECOMMENDATIONS	43
Refere	ences	44
Apper	ndix	53

ABSTRACT

'Entrepreneurship' and 'economic development' are the key factors of any developed or developing country in the world. This paper examined policy disparities across countries at different stages of development, focusing on the outcomes and the governments' critical role in guiding entrepreneurial decisions. A comparative qualitative methodology was adopted, selecting India, an emerging economy with a significant service and manufacturing sector, and Ireland, European Union nation known for its robust business support infrastructure in information and communication technology (ICT) and finance. Employing a qualitative methodology, this study conducted semi-structured interviews with eight entrepreneurs and small-scale business owners from India and Ireland, focusing on how governmental policies during the COVID-19 pandemic had influenced their businesses, with an analysis grounded in international policy impact criteria. Interviews with the entrepreneurs and small-scale business owners from these nations provided insights into the impact of government policies on business during the COVID-19 pandemic.

The comparison of India and Ireland showcases diverse entrepreneurial ecosystems and varying policy responses, offering a comprehensive understanding of the international entrepreneurial landscape. Findings suggest that policy decisions are paramount not only for individual entrepreneurial success but are also critical for the collective prosperity of a nation. Recommendations emphasize the need for adept policy-making that equips small-scale businesses to thrive, fostering a favourable environment for new ventures and investments that contribute to economic resilience and growth. This research examines how government policies affected entrepreneurship during the COVID-19 pandemic. It compares the situation in Ireland and India, finding that Ireland's strong business support and proentrepreneurship policies helped mitigate the crisis' impact on small businesses. Meanwhile, financial aid, streamlined regulations, and targeted support programs fostered an entrepreneurial spirit even in difficult times.

1. INTRODUCTION

The research study focuses on the critical issue of how government policies influence entrepreneurial activities in the case of economic crises. This research paper makes a comparative analysis of entrepreneurs and their experience about running their businesses in India and Ireland. The main aim of the study was to assess the impact of government policies on developing entrepreneurial attitude, contribution towards economic development, to identify opportunities and challenges faced by entrepreneurs and to evaluate the overall economic performance and crisis-induced policies. It will provide an outline of policies and reforms introduced by respective government during Covid-19 pandemic and how it influenced the entrepreneurs. This paper mainly focused on the fostering entrepreneurial mindset through government policies which will guide both the policymakers and entrepreneurs in future at the time of economic crisis.

Governments' policies are crucial in encouraging entrepreneurship, as shown by the COVID-19 outbreak that greatly injured world economies. According to Vatavu *et al.* (2022), the government policy has had a positive impact on entrepreneurship but research findings were region-specific, mostly G8 countries and although at times crucial the research did not find out what exact part of the policy influences during global economic crisis.

This research sets out to fill this very gap by studying the efforts that the Governments of Ireland and India have made toward boosting entrepreneurship amid the pandemic, and the impact caused on their economies. Particularly, it will look into the role of government policies in influencing entrepreneurial attitude, economic growth contribution during times of crisis, opportunity and challenge identification for entrepreneurs, and evaluation of overall economic performance with respect to policies related to crisis management (Newman, Obschonka and Block, 2022).

The impact of government policies on entrepreneurship and economic development is very important to be studied against the rapidly changing economic landscape today. The world has suffered from the pandemic caused by COVID-19, and therefore, the understanding of the role of government policies in putting

forward a favourable environment for entrepreneurial activities has become of essence. The study by Benlaria *et al.* (2023) explains the role of governments support which play a very vital role in shaping the entrepreneurial landscape of an economy towards economic growth and development. Most especially in times of crisis, supportive and adaptive government policies are needed to help small-scale businesses and entrepreneurial ventures survive and thrive. It is within this comparative analysis between India and Ireland that some valuable lessons have been found within the divergent entrepreneurial ecosystems and diverging policy responses to an economic crisis. Indeed, how different governments fare in dealing with the challenges brought about by the pandemic may also serve as a benchmark to steer the course of policymakers toward the design of proper strategies to ensure entrepreneurial viability during periods of economic crisis and long-term economic growth.

The present research once more underlines the necessity for international cooperation and exchange of best practices in policy-making, seeking to improve entrepreneurial ecosystems across a wide variety of socio-economic contexts. Drawing on the experiences of India and Ireland, this paper proposes specific policy recommendations for both countries and beyond. These recommendations focus on the role of tailored policy measures in fostering an enabling environment for entrepreneurship and innovation during periods of crisis.

The COVID-19 pandemic has resulted in an unprecedented global economic crisis that has revealed the fragility of different world economies. The major cause of resilience in countries like Ireland was due to their specific relation with multinational enterprises, especially in the pharmaceutical and technology fields (Polyak 2022). This experience renews interest in the role of government policy and its effect on such economic catalysts. Thus, similar challenges were relevant to developing nations in which entrepreneurship became even more central to socio-economic stability. The promotion of entrepreneurship is very much intertwined with the goals of sustainable development and the pursuit of economic recovery post-pandemic (Abdelkafi and Refas, 2021). Yet, its role stays so vague and dramatically differential

between contexts and economic development stages that it underscores the need for these policy approaches to be nuanced.

The structure of the paper is as follows:

- Section 2 presents literature review on the relationship between government policy and entrepreneurship at the times of economic crises and its impact on the business.
- Section 3 states the research question and its objectives and validation for carrying out comparative analysis between Ireland and India.
- Section 4 describes the method used to carry out this comparative study between India and Ireland.
- Section 5 presents findings based on entrepreneurial ecosystems and policy responses of India and Ireland
- Section 6 presents the significance of these results by discussing main outcomes, implications and policy recommendations.
- Finally, the last section is a summary of the research findings with areas for further studies.

2. LITERATURE REVIEW

2.1. Impact of the Covid-19 Pandemic on Entrepreneurship

All the industries irrespective of the size were affected by Covid-19 pandemic. Though tourism and the aviation sector were seriously affected due to restrictions, the IT Sector showed innovation and resulted in an increased number of start-ups in developing economies like India (Kumar *et al.*, 2020). However, the study by Fikri and Newman (2024) proposed that throughout the covid period, the start-up rate was not evenly distributed. There is a possibility of starting more unconventional business-like online grocery shops as others have gone high, but if we look into various sectors and regions for overall performance, it is inconsistent concerning business creation across all of United States.

This change in consumer demand and positive and negative effects on markets to a large extent open up several opportunities that were hastened by the pandemic. Entrepreneurs have faced difficulties during this pandemic while they showed their ability to adapt and innovate. However, during difficult economic times, government policies have been vital in promoting the spirit of entrepreneurship especially in countries like Ireland and India (Krishnan et al. 2022; OECD, 2021). Such dynamics can be explained using both Institutional Theory and Entrepreneurial Ecosystem Theory which stress on the role of state in determining entrepreneurial process as well as its environment.

According to Vadeboncoeur and Jennifer (2023), institutional theory looks at the way government policies, laws, regulations, and social norms determine economic activities and organizational behaviour. The theory argues that such institutional factors create guidelines for organizations and people on how to carry out their mandate in a particular context. This means that even when they are not the most efficient or effective from a technical perspective, organizations may adopt certain structures and practices which appear legitimate or acceptable in their institutional environment.

Institutional Theory elaborates on the influence of government policies and institutional environments that affect economic activities and organizational behaviour. During COVID-19 pandemic the Irish government carried out several measures aimed at assisting entrepreneurs such as financial assistance, tax favourites and advising. For instance, the Enterprise Support Grant (Department of Social Protection, 2023) provided financial assistance to businesses negatively affected by COVID-19 pandemic to adapt and survive financially. Similarly, the Startup India initiative (2024) implemented by the Indian government offered funding, incubation centres and lessening of rules for new companies. This was further supported by the emergency credit line guarantee scheme to facilitate cash flow to small and medium enterprises during this period of hardship. Thus, these institutional supports were of immense help in sustaining entrepreneurial activities during this period of crisis.

The theory of an Entrepreneurial Ecosystem helps identify that complex system of actors, institutions, and processes that create an enabling environment for entrepreneurship. A well-established entrepreneurial ecosystem in Ireland is facilitated by government interventions like Enterprise Ireland which are essential for establishing start-ups and encouraging innovations (Department of trade enterprise and employment, 2024; Low, 2005). The Sustaining Enterprise Fund established by Enterprise Ireland provides financial resources to companies during these trying times so that they can have an opportunity to regain stability and expand their activities (Enterprise Ireland, 2024). The importance given to digital infrastructure in India through initiatives such as Digital India considerably leveraged the entrepreneurial ecosystem and encouraged the fast growth of startups like Byju's, an Ed-tech company that underwent immense expansion as a result of the tectonic shift towards online learning when the schools closed down (Dingliwal and Sharma, 2023). The COVID-19 outbreak has highlighted the role that the governments play in coming up with regulations and creating a conducive environment for business so as to help and promote entrepreneurship during times of economic downturn.

Government policies and incentives aimed at lowering barriers have been critical enablers of entrepreneurial development. As an example, the Future Growth Loan Scheme in Ireland provided long-term financing that permitted firms to take risks on innovations and expansion plans during hard economic times (AIB 2024). Likewise, there was a boost to small scale business undertakings in remote parts of India through microfinance programs like the Pradhan Mantri MUDRA Yojana (Mishra, 2024). These interventions focused on shaping up different ways of thinking about entrepreneurship so as to create an atmosphere where risky but thrive. the fruitful investments Additionally, Behavioural Theory Entrepreneurship investigates how outside forces such as government policies influence individual behaviour and decision making with regard to initiating new ventures or making further investments in existing ones.

The supportive policies in Ireland and India reduced perceived risks for entrepreneurs and encouraged more people to start businesses despite the economic downturn (Pandey et al., 2003). The government policies were more focused on self-employment and digital entrepreneurship, which the government was gearing up with different resources and measures. The impact of supportive government policies on entrepreneurial decision-making and market adaptation is exemplified by the proliferation of online grocery providers such as BigBasket (Misra et al., 2022). The interactions among the government's policies, entrepreneurial ecosystems, and individual entrepreneurial behaviour are strongly interwoven ways in which the COVID-19 pandemic influences entrepreneurship in Ireland and India. While posing huge challenges, the pandemic also became an impulse for entrepreneurial innovation and growth opportunities. In both cases, one can trace the role played by government policy in terms of nurturing the entrepreneurial mindset with very necessary supports and resources. The present research underlines this in an analysis of experiences: tailored policy measures are of key importance to the promotion of entrepreneurship and economic resilience during crises.

2.2. Entrepreneurship and the Economy

The 21st century with its exalting rise to the level of adoption by governments and economies worldwide to the principles of a circular economy has created an uncommon level of interest in respect to sustainability. An uncommon level of interest means giving way to new vistas and possibilities that, in turn, have equally spurred new chances and innovation through entrepreneurship. The circular economy principles focus on waste reduction, reuse of materials, and regeneration of natural systems relating to the creation of the frame within which the entrepreneurial venture can thrive (Nuthalapati and Nuthalapati, 2021). As emphasized by Neumeyer, Ashton and Dentchev (2020), these principles have revolutionarily changed entrepreneurial landscapes with the creation of sustainable business models concerned with environmental stewardship and, most importantly, economic growth.

As Demirel and Danisman (2019) delineate, one study emphasizes that there is an overall net effect of the entrepreneurial demand in developed and developing economies due to the circular economy. In the developing economy of India, government policies under the aegis of the Startup India program have been inculcated with these attributes, orchestrating new innovative ventures which are centred around sustainability (Das and Rangarajan, 2020). Initiatives like Enterprise Ireland and Startup India have enabled platforms for the entrepreneurs to generate businesses that are going to contribute towards this economic growth and, at the same time, are going to be environmentally sustainable. For instance, some launched activities under Startup India have focused on developing and creating green, sustainable products and services, including from green packaging solutions to renewable energy technologies (Nayak and Sahoo, 2021). Emergent start-up ecosystems are nurturing the rapidly growing sustainable entrepreneurship.

The study by Rao and Nutthalapati (2021) found that The digital platforms have been instrumental for enhancing innovations of SMEs in markets of fertilized grounds. For instance, a startup like Swiggy and Zomato has changed the

dimension of food delivery in alignment with the broader goals of sustainable development in new markets, such as in India. These companies have integrated digital technologies to plan optimal delivery routes, diminish food waste, and promote the use of biodegradable packaging (Meenakshi and Sinha, 2019). They have demonstrated compatibility between economic success and environmental responsibility through the business model integration of sustainability. Entrepreneurship represents a means by which economic advancement can be achieved, as it can drive changes in investments, income distribution, innovation, and productivity.

In fact, Tesla and SpaceX are great examples to show to which extent entrepreneurial activity helps in contributing to economic development in innovation (Ingram, 2018). The ventures undertaken by Elon Musk define new areas of transportation and space exploration while its spinoffs are creating high-skilled employment and furthering economic development. In particular, Tesla electric cars transformed the automotive industry and facilitated the growth of new economies, such as those related to battery technology and sources of renewable energy (Muegge and Reid, 2019; Ingram, 2018). Similarly, cost-cutting and pricing innovation in the space industry has enabled the commercialization of space exploration, satellite launches, and inter-planetary missions. Literature shows that digital entrepreneurial activities are a driving force behind economic development in a rural economy.

For instance, in India, the Digital India initiative has empowered entrepreneurs with the ability to leverage digital technology for economic development, even in the most remote parts of the country (Sindakis and Showkat, 2024). The engagement of rural entrepreneurs in the global economy, by accessing the market, digital infrastructure, and resources available, has had a series of resultant outputs: generation of local employment and empowerment. These, for example, are the rural artisans who can now easily sell their products through e-commerce means and even extend the reach to sell their products on a global platform (Mehra, Mathur and Tripathi, 2019). Impact assessment of government policies

during a time of crisis is, therefore, a key source of insight into societal views and regulatory environments under which entrepreneurial responses can be shaped and will, in turn, affect regional resilience.

Policy Comparison: Ireland and India:

The policies of the government in Ireland emphasize a reduction in regulatory burdens and the provision of financial support to entrepreneurs, enabling them to keep adapting and innovating in the unfolding face of economic challenges (Lin, Shen and Chou, 2010). On the other hand, India's policies lay much emphasis on digital empowerment and infrastructure development that supports remote working and online business, particularly in times of crisis (Singh, Venkataramani, and Ambarkhane, 2021). These various strategies provide key lessons in utilizing government policies to foster entrepreneurship, which is a driving strategy of any significant recovery effort. Realization of how the dynamics of government policy, societal attitudes, and entrepreneurial enterprise come into contact amidst turmoil helps shed light on why an entrepreneur-driven development process and economic resilience are therefore able to work so powerfully at a global level (Tuszynski and Stansel, 2018). This may certainly offer great insights for the formulation of sound options and strategies toward guiding entrepreneurship potentials in harnessing the same for even more effective economic challenges post-any crisis situation. Governments can increase regional resilience, foster sustainable development, and lead to long-term economic growth in an entrepreneurial enabling environment.

In the conclusion, the 21st century has experienced a confluence between entrepreneurship and a circular economy with innovative sustainable development. Here, the efforts of the government, digital platforms, and enabling policies become boosted in providing entrepreneurs with a framework for coming up with enterprises to drive economic growth alongside environmental protection. On the other hand, with the continuous development of the global economy, one of the key components answering tomorrow's challenges and ensuring long-term

vibrant development includes creating an environment that supports entrepreneurial pursuits.

2.3. Role of Government in Promoting Entrepreneurship

Most start-ups related to business in India often die out because of insufficient sales and marketing strategies and, at times, a lack of funding, argues Goswami, Murti and Dwivedi (2023). In spite of this fact, the Government of India still encourages entrepreneurship through policies and current campaigns. A study by Gan, Zhang and Zhang (2023) can further elucidate the government's inclination towards supporting entrepreneurship, wherein the authors consider how one policy of economic reform, adopted in China back in the 1970s, enabled it to emerge as the world's prominent manufacturer. This 'grow SMEs' policy supported SMEs in becoming the overall positive economic development of China.

Government involvement in promoting entrepreneurship is critical for a host of reasons. One primary motive has to do with the alleviation of poverty and improvement in the quality of life. By having policies that especially prop up poor entrepreneurs, the governments are creating an opportunity for job creation to give people the means to be independent and have a livelihood. Newman *et al.* (2022) represent, in their conceptual model, crisis navigation by entrepreneurs during crises such as significant events like the COVID-19 pandemic, which further reveals their resilience and adaptability of entrepreneurial ventures in challenging circumstances. Vatavu *et al.* (2022) go further to establish that the government policies encouraging entrepreneurial activities have seen a positive impact on GDP in G8 countries, hence establishing the broader economic impact of such policies.

The government can be expected to influence the entrepreneurial ecosystem through its financial, infrastructure-related, and regulative support. The study by Lee (2019) explores that government support is one of the most important things

for creating the conditions under which entrepreneurship may thrive. Normally, the governments share the financial burden of start-ups by offering financial incentives, tax breaks, grants, and subsidized loans so that the entrepreneurs may focus on growth and innovation. The infrastructure is an impetus to innovation because it provides the entrepreneur with all the necessary resources and networks for business growth through technology parks and business incubators, as noted by Williams (2013).

Clearly, government support is needed if entrepreneurship is going to progress, but a fine balance between state involvement and market-driven innovation should be maintained. Audretsch, Keilbach and Lehmann (2006) argue that when governments become too controlling, they tend to impede on entrepreneurial spirit and the response rate of entrepreneurs towards market dynamics. In this regard, well-crafted policies should be both implemented and constructed so as to create a conducive environment for development while avoiding those regulations likely going against innovativeness and growth (Cadar and Badulescu, 2015). The intersection between public policy theory with respect to entrepreneurship has also been demonstrated in research done by Khyareh and Amini (2021), which highlights how governmental interference can affect or promote the behaviour of entrepreneurs thereby stimulating economic growth. For instance, some countries such as India have adopted similar policies where by in addition to reducing risks associated with entrepreneurship, they have promoted some businesses (Jahanshahi et al., 2011). That means public policy can be argued to be a major driver of entrepreneurial mindsets and behaviours.

The interdependent relationship between digital entrepreneurship and economic development, more particularly in rural areas, goes to the power of government policies in driving not just the entrepreneurial response but also its positive fostering towards economic resilience. Programs like Digital India have enabled entrepreneurs to leverage digital technologies in ways that are productive for economic growth and even job creation in some of the remotest areas (Tandon et al., 2020). By providing access to digital infrastructure and the

promotion of digital literacy, governments could give innovating entrepreneurs the wherewithal to grow their businesses and create a more inclusive economy.

In short, it is the duty of governments to promote entrepreneurship through creating an enabling environment for aspiring entrepreneurs with corresponding support mechanisms. The interaction among the policy of government, attitude of society, and activities of entrepreneurship has to be critical in illuminating the formulation of policies that ensure sustainable economic development. As such, governments have to achieve a delicate balance between support and regulation so that this entrepreneurship thrives to sustain economic growth, generate employment, and improve living standards. It is through targeted interventions and a commitment to creating an enabling environment that governments are able to unlock the potential of entrepreneurs and drive long-term economic prosperity.

2.4. Entrepreneurship and Financial Independence

Financial independence is mostly derived from entrepreneurship because it entails creating, owning, and managing private businesses. This kind of ventures are not limited by salaries or fixed pay; they are therefore much more rewarding compared to formal employment options (Blanchflower and Oswald, 1998). This possibility is framed by Economic Agency Theory, which suggests that entrepreneurship authorizes people to become masters of their economic lives (Lowrey, 2003). This is not merely personal but social empowerment, too, since entrepreneurial activities drive economic progress by creating jobs, stimulating local economies, and generating taxable revenue streams.

By creating jobs and reducing unemployment, entrepreneurs contribute to economic growth and increase community stability. Most of such business ideas require a supply chain of goods and services needed for their completion thereby leading to well established new businesses that provide support for local economy due to their importance in terms of stimulating them. According to

Shafiu, Manaf and Muslim (2020), the taxable income generated by entrepreneurial activities provides revenue that funds the government's financial resources, enabling the financing of public services and infrastructure, thereby strengthening the economy, a trend observed in many developing countries (Hamdan, 2019).

Additionally, the acquisition of assets that are highly valued, like real estate or intellectual properties is attainable by entrepreneurial businesses. This enables them to attain a better financial position due to the security provided by these assets and the potential for their appreciation in future. (Alvarez and Barney, 2014). This Resource-Based Theory entails that irreplaceable resources characterized by rarity or value can provide sustained competitive advantage to any firm. This involves both tangible assets, such as physical and financial resources and intangible assets, such as knowledge, capabilities, and organizational processes. Grant (1991) indicates that companies in a position to utilise the rare and valuable resources will outperform their competitors in the marketplace. Also, the study conducted by Alvarez and Busenitz (2001) affirmed this notion that entrepreneurial ventures normally prosper because entrepreneurs make use of unique assets like creative ideas as well as innovative business models which hence gives them a competitive edge over rivals within the market.

Entrepreneurship can be defined as pursuing a mindset that has the ability and characters of a risk-taker, an innovator, and a resourceful individual to sail through uncertain situations of any business environment, which leads to financial independence (Kolvereid and Isaksen, 2012). Thus, calculated risk-taking can give entrepreneurs a chance to challenge traditional markets, launch novel offerings and produce worth in manners that are frequently disregarded by current businesses. As Matenda and Sibanda (2023) mention, this form of entrepreneurial innovativeness is essential for both the entrepreneur himself and general economic growth. By introducing competition, it pushes industries to continuously improve and better themselves.

The entrepreneurial initiative is an innovation catalyst, lying at the very root of economic development. Innovations that entrepreneurs launch might change entire industries in a quite fundamental way, enhancing the efficiency of what is being done and meeting unsatisfied needs in markets. According to McClure and Thomas (2021), creative destruction, a concept termed by the Austrian economist Joseph Schumpeter, is important for economic progress. This process substitutes obsolete practices with more effective solutions, leading to improved products and services and advancing society in general (Abrams, 2007). Also, the job market can benefit from entrepreneurial activities since newly born businesses usually create different types of jobs, from simple ones to special ones, thus shaping a dynamic labour market adjusted to different skill levels and professional aspirations. As Matenda and Sibanda (2023) explain, entrepreneurial initiatives can induce competition that forces existing businesses to innovate and improve their products, thereby benefiting consumers through higher quality products at lower prices.

In a nutshell, entrepreneurship paves the way toward financial independence, growth of wealth, creation of jobs, and generation of taxable income. Entrepreneurship is therefore a means to financial independence and social development, as it fuels continued economic success. Through innovation, entrepreneurial firms not only acquire advantages but also help to better society by fostering competition and stimulating the job market (Gashenko et al., 2016). The economic productivity of an entire nation relies heavily on the effectiveness of business start-ups, as they promote diverse economic activities. Financial sustainability and expanded economies require that entrepreneurship begin with an intrinsic vision which has persistence across generations. Without a doubt, entrepreneurship goes beyond personal gains since it creates a basis for addressing social concerns and resilience against downfall through invention. Entrepreneurship as a catalyst for economic growth creates a vibrant dynamic society by giving people the power to determine their own destinies economically. This process ensures individuals' ownership over their destinies, propelling them towards development.

2.5. Entrepreneurship in Developing and Developed Nations

The significant difference between entrepreneurship in developed and developing countries can be attributed to their varying economic characteristics and resource availabilities. Gouvea, Kapelianis and Montoya (2018) indicate that, entrepreneurs in developing economies face numerous problems like terrible infrastructure, absence of government support and limited access to capital. Therefore, entrepreneurs are constantly innovating by coming up with novel ways of utilizing available resources effectively to their respective environments despite the challenges. These solutions often revolve around primary services especially health care, education etc that contribute directly towards socioeconomic growth. For instance, in this area, they usually come out with innovative products at a low cost that are supposed to have a great impact, demonstrating grassroots innovation and durability (Shkabatur, Bar-El and Schwartz, 2022)

In contrast, entrepreneurs residing in developed countries are privileged to be backed by more solid systems of support that feature aspects such as access to venture capital, advanced technology, comprehensive governmental and institutional backup among others (Engel, 2015). The entrepreneurial terrain in these areas is characterized by an emphasis on disruptive innovations and technology-based businesses. Developed countries normally have well-defined infrastructure, high availability of trained manpower as well as conducive legal frameworks for business expansion. The aim for many entrepreneurs found here is global market penetration through the application of modern technologies to create new products and services (Williams, 2013). As a result, there are several types of entrepreneurships which differ necessarily with respect to whether they pioneer innovative ideas or just exist to take advantage of the market opportunities created by other existing players.

While both settings aim for economic growth and job creation, the nature of entrepreneurship reflects the unique challenges and opportunities present in each environment. Remarkably, the study by Koch and Caradonna (2006) indicates developing nations demonstrate a significant degree of resilience and

inventiveness at the community level, tackling urgent social problems with little resources. On the contrary, developed countries prioritize high-tech innovations and possess resources to back large-scale, high-risk initiatives that have the potential to transform industries (Williams, 2013). In order to bridge these two contexts, customized policies that support entrepreneurs according to their specific surroundings are necessary. This may help in developing an inclusive entrepreneurial ecosystem which is globally connected thereby enhancing knowledge sharing and collaboration between developing and developed regions.

The connectivity between entrepreneurship, crisis situations such as pandemics, and government regulations underscores the significance of flexible strategies for crisis management while promoting innovation and sustainable economic development. In many cases, crises are used as motivation by entrepreneurs to innovate (Boudreaux, Jha and Escaleras, 2023). For instance, it is during the COVID-19 pandemic that there were numerous health-techs, online offices and shopping sites for those who saw an opportunity. This crisis-driven innovation demonstrates the critical role that entrepreneurs play in adapting to and overcoming unforeseen challenges.

Government regulations and policies are pivotal in shaping the entrepreneurial landscape during such crises. The study by Kuckertz *et al.* (2020) states that flexible and supportive regulations can enable entrepreneurs to quickly adapt and scale their innovations, contributing to societal resilience and economic recovery. On the other hand, tight regulatory measures may dampen creativity and prevent start-ups from reacting effectively. Hence, governments should aim at balancing between all needed supervision and allowing enough freedom for new businesses to flourish.

Assessing the potential of entrepreneurial ventures in partnership with governmental initiatives is essential for shaping a resilient and thriving society. The private and public sectors working together, will create innovative solutions for both short-term and long-term problems (Gigauri and Damenia, 2020). For

instance, public-private partnerships facilitate the introduction of contemporary technologies in health services delivery systems, educational institutions and infrastructure, thereby enhancing the overall resilience of societies. According to Lima *et al.* (2021), such collaborations play a vital role during periods of crisis, as they foster sustainable economic growth and innovation.

Finally, it is the economic environment of developing and developed countries that define the unique challenges and opportunities entrepreneurship within them. On one hand, the developing countries display an amazing adaptive quality as well as inventiveness from below, on the other hand, there are highly sophisticated technologies available in industrialized states which support entrepreneurship. Policies that are specific to each country's context can be used in bridging this divide between these two groups of nations so that they can create an inclusive entrepreneurship environment that is globally connected. In addition, the interplay of entrepreneurship, crises and regulatory frameworks by governments necessitates the need for adaptive strategies aimed at promoting innovation and economic sustainability. A society can harness both private and public sector strengths to build resilience against future adversities.

3. RESEARCH QUESTION

This proposed study is mainly focused on the following question:

What are the initiatives taken by the Governments of Ireland and India in promoting entrepreneurship during the pandemic and how did they benefit the individual and nations' economies?

Research Objective:

The proposed study has the following objectives:

- Examine the impact of government policies in fostering the entrepreneurial attitude among their citizens in Ireland and India
- Comprehend how these policies would contribute to the economic development during crisis in both nation
- Identifying the opportunities and challenges faced by entrepreneurs in Ireland and India due to the policies formulated during pandemic
- Evaluate the performance of the Irish and Indian economy in respect to the policies initiated during the pandemic.
- Compare and contrast the pros and cons of the policy implemented by Ireland and India during pandemic along with the adaptability of the initiative in both the countries.
- Provide possible recommendation for policymakers based on the finding and aiming to enhance the government's policy to promote entrepreneurship in context of similar crisis.

The next chapter will discuss the methodology used to achieve the objectives and the research question.

4. METHODOLOGY

4.1. Philosophy

The interpretivist paradigm provides the theoretical foundation for this research. According to Kratochwil (2008), this philosophy popularly referred to as constructivism proposes that reality is a social construct and can only be understood from subjective experience and interpretations of individuals. This particular methodology is key for this study, which aims at examining how government policy may encourage entrepreneurship during economic crises by comparing Ireland and India. The study adopts an interpretivist perspective in order to grasp the nuanced and complex ways entrepreneurs see and react to government policies in different social economic environments (Laing, van Stel and Storey, 2021).

The interpretive philosophy fits well with the aim of this research as it gives importance to the views of individual entrepreneurs. This is not only about identifying the policies adopted, but also grasping how these policies are seen and experienced by those that suffered (Martínez, 2009). This approach enables a deeper examination of contextual and situational problems affecting entrepreneurial conduct which are necessary for making comparison between two different nations such as Ireland and India.

4.1.1. Ontology and Epistemology

In this study, ontology signifies the reality of government policies and their effects on entrepreneurship. The ontological postulation in this case is that government policies are concrete actions that have an effect on the entrepreneur eco-system (Jahanshahi *et al.*, 2011). These policies can either promote or hinder entrepreneurial activities, depending on how they are designed and implemented.

According to epistemology, knowledge about the impact of government policies on entrepreneurship can be acquired through the experiences and perspectives of entrepreneurs. The purpose of the research is to get a complete

and contextual understanding of these experiences. The study by Javadian *et al.* (2020) indicates that this is achieved through qualitative methods such as in-depth interviews, which allow for the collection of rich, descriptive data that can provide insights into the entrepreneurs' perspectives.

The research approach is influenced by these ontological and epistemological beliefs as they determine the picking of qualitative techniques that emphasize profoundness instead of extensiveness. The detailed stories given by entrepreneurs in this research will reveal the complicated relationship between policies and their responses from entrepreneurs.

4.1.2. Justification for Interpretivist Approach

The choice of interpretivism is justified by the need to explore the subjective dimensions of policy impacts. Numerous scholars have utilized valid theories and models such as quantitative objects, which help them estimate trends of certain variables. Despite being important aspects targeting policy impacts among individual business owners, Williams (2013) also noted that they fail to capture lived stories from ordinary people in society. Interpretivism, in contrast, offers an opportunity for a closer look at these experiences thus providing deeper and fuller insights into the subjects being researched (Brattström and Wennberg, 2022).

Contextual variations and similarities can be examined using interpretivist approach represented by such comparative study (Pickard, 2013). Ireland & India differ in their economic structures, cultural contexts, and policy environments. A more detailed context-sensitive analysis can therefore be provided by using the interpretivist approach in this research.

4.1.3. Influence on Similar Studies

There are a couple of studies that have successfully employed interpretivism to explore similar research questions. For instance, Yao, Cahn and Yao's (2024) study provides an interpretivist methodology for investigating consumer behaviours in physical and digital environments. Qualitative research comprises a

good strategy for understanding different contexts and people's perceptions or interrelationships towards those places. These insights could significantly improve phygital marketing strategies as evidenced by Liu *et al.*'s (2018) study which used qualitative interviews to gauge the role of government funding in fostering partnerships between educational institutions, research bodies, and industries across China's solar photovoltaic sector. The concept of behavioural additionality was used to examine changes in research behaviour due to government funding, offering a nuanced understanding of its impact.

These studies demonstrate the value of an interpretivist approach in comparative research, particularly when exploring complex and context-dependent phenomena such as the impact of government policy on entrepreneurship. By focusing on the subjective experiences of entrepreneurs, these studies were able to provide detailed insights that would not have been possible with a purely quantitative approach.

4.2. Research Design

A comparative research design was employed in analysing the government policies and initiatives taken by Ireland and India in promoting entrepreneurship. The methodology was based on the mixed qualitative study which was in line with the research objective. The study made use of academic journals as well as in-depth interviews.

4.2.1. Comparative Research Design

This study employs a comparative research design to analyse the influence of government policies on cultivating an entrepreneurial mindset during economic crises, specifically comparing Ireland and India. This comparative research approach enables the investigation and comparison of the distinct methods employed by Ireland and India to cultivate entrepreneurship (Low, 2005). It is particularly useful for understanding how diverse socio-economic and political contexts impact policy effectiveness. The choice of a comparative design aligns

with the research objective of evaluating the impact of government policies on entrepreneurship. The study compares two different situations in order to find best practices and possible areas where policy could be improved.

4.2.2. Mixed Qualitative Approach

A mixed qualitative methodology was used to explore the research questions comprehensively. It consists of different qualitative techniques such as in-depth interviews, review of secondary data sources, and a policy analysis. The strengths of employing this multi-method design lie in its ability to provide a more holistic and nuanced understanding of the research problem (Abutabenjeh and Jaradat, 2018). Qualitative design is particularly appropriate here because it allows for an exploration at micro levels as well as consideration of entrepreneurs' lived experiences within their contexts.

- In-depth Interviews: All the primary data are collected from the business owners of small and medium enterprises located in both India and Ireland. Interviews were conducted using Microsoft Teams or Zoom according to the interviewees and recordings were done with consent from the participants. Such semi-structured format of interviews ensured that questions relevant to the experiences and perceptions of interviewees were appropriate.
- Policy Analysis: The study involved analysing government reports, policy documents, and economic indicators. There were various categories of data such as that found in government reports, policy documents, economic indicators etc. The second set of data provided context for interpreting the interviews' primary information. The study by Koster and Rai (2008) applied a similar approach to examine the relationship between the entrepreneurship and economic development in developing and developed nations. This study focused on policy analysis helped in understanding the specific measures taken by the governments of Ireland and India to promote entrepreneurship during the economic crisis.

■ Content Analysis: This study was based on a content analysis of scholarly journals, third party survey results, and policy cases studies. For the purposes of this study it helped in recognizing key concepts and recurring themes (Roller and Lavrakas, 2015). A comprehensive interpretation was made by analysing primary data collected through interviews with the abovementioned secondary information.

4.2.3. Integration of the data sources

The integration of academic literature and primary interview data was accomplished through thematic analysis. This involved coding of information and identifying the important themes arising from both primary and secondary sources. The criteria for selecting secondary sources included their relevance to the research objectives, credibility and recency (Roller and Lavrakas, 2015). For instance, government reports, policy documents, and economic forecasts were prioritized because they directly concerned the study's focus on policy impacts. Entrepreneurs participated shared their experiences and also views towards governmental support during the crisis period through an extensive study of interviews with them.

4.2.4. Validity of the data

Maguire and Delahunt (2017) emphasise the importance of validating the data in order to interpret the data accurately which aids to prove or disprove the research objective. The author employed a thematic analysis to strengthen the validity of the findings from the data collected through interviews, policy documents, and academic literature and the content were compared and analysed in detail. The results were therefore validated and possible biases were eliminated. Furthermore, in order to guarantee the accuracy of the information, writer respected a standard interview protocol and wrote down all interview answers word by word. Moreover, secondary data from reliable sources was added to enhance the general trustworthiness of the study.

4.2.5. Addressing Bias

Bias was minimized through careful selection of participants and rigorous data analysis. The criterion for choosing participants was whether they had set up business enterprises or supervised them when there were economic recessions so that their feedback would be valuable. There was consistent use of coding and re-evaluation within theme analysis among the researchers to eradicate any resultant biases. The time frame of this research was mainly around pandemic years, directing its objectives toward that aspect.

4.3. Participants and Sampling Size

This Qualitative analysis included six to eight small as well as medium-sized business owners from Ireland and India who either started or endured much of their operations during pandemic. Bullard (2023) elaborates purposive sampling as a non-probability sampling technique in which participants are selected on voluntary basis considering certain factors or qualities related to the study. Unlike random sampling, purposive sampling targets those individuals who can provide relevant information about specific areas of focus within research (Saunders, Thornhill and Lewis, 2019). Purposive sampling focused on selecting customers most impacted by coronavirus pandemic or recession directly or indirectly. These entrepreneurs were ready to freely share their personal views and opinions on how they manage these never seen before difficulties that business people face. Thus, this was an adequate representation of all entrepreneurs drastically influenced by wide-ranging implications of outbreak.

In this study, five Indian entrepreneurs participated, out of which three were males and other two were females. Each participant had begun their businesses before the onset of the pandemic and had made strategic modifications during the crisis. On the other hand, in Ireland, the sample comprised three male entrepreneurs whose businesses had been affected by the pandemic and who utilized government assistance programs for their continuity. This study confined

itself to only a few cases because qualitative research emphasizes on a thorough investigation rather than wide-ranging generalization

Details of the participants:

Participants	Name of the Company	Country	Sector	Did their business affect by Covid-19 Pandemic?	Did they benefit by Govt. initiative/policies?
Participant 1	Orahi	India	Information Technology	Yes	Yes
Participant 2	Eva Shipping	India	Transportation & Logictics	Yes	Partially
Participant 3	Skill Live (Formally Yeh China)	India	Education	Yes	No
Participant 4	RA Energy	India	Renewable Energy	Yes	Yes
Participant 5	Ark Consultancy	India	Aerospace & Defence	Yes	Yes
Participant 6	Ecco Spray Limited	Ireland	Medical Technology	Yes	No
Participant 7	Connect Promotions	Ireland	Consumer Goods	Yes	Yes
Participant 8	Oracawise Al	Ireland	Artificial Intelligence	No	No

4.4. Data Collection Method

Saunders et al. (2019) define semi-structured interviews as non-standardised and often referred to as qualitative research interviews. The researcher has a list of themes and sometimes key questions, but they do not use them in a structured way. Pilot data collection was conducted with one entrepreneur from each country. The inputs and feedback gathered during this phase helped to refine and frame the open-ended questions used in the semi-structured interviews for the main study. The researcher may employ some questions in certain interviews and

different questions in others. Primary data was collected through in-depth, oneon-one interviews conducted using Microsoft Teams or Zoom Video conferencing software. Participants' consent was obtained prior to recording the interviews. The interviews followed a non-standardised, semi-structured format to accommodate the diverse experiences exposed during the COVID-19 pandemic. In addition to the primary information obtained from the first-hand interview, this research relied on other sources like reports by governments, policy documents, various surveys carried out by agencies, alongside economic indicators and case studies as well as experiments that focus on entrepreneurial mindset. The study by Santoso and Fianto (2022) adopted similar methodology where they employed exploratory qualitative research model, by analysing the economic and financial theories, integrating research on economic patterns in creative industries. These multiple data sources all contributed towards the assessment of the research objectives. These interviews aimed at understanding what entrepreneurs think about the challenges and opportunities they got from COVID-19 pandemic when running their businesses and also how they experienced support from government. In cases where interviews could not be conducted, respondents were given openended questions (attached Appendix - I) through email or other messaging platforms.

Limitations:

This study considers the limitation of using a qualitative research methodology. Although the mixed-method approach might have potentially retained both qualitative and quantitative data collection and analysis for stronger and more comprehensive insights, this study focused only on qualitative data collection and thematic analysis due to the time constraint. Future research work would be enriched if a mixed-method approach were used, creating a holistic understanding of the research topic.

4.5. Data Analysis

According to Lapadat and Lindsay (1999), data analysis is a process that encompasses interpreting as well as systematically examining it in order to extract

important insights. The main purpose of data analysis is accurate defining of data through identifying patterns, trends and relationships within the collected information (Saunders et al., 2019). In this study, thematic analysis was conducted on data from both interviews and secondary sources such as government reports and academic literature. Thematic analysis from both interviews as well as secondary sources such as government reports & academic literature was conducted. Interviews were transformed into word-to-word transcripts, and then the data was coded and categorized systematically. Thematic analysis method has been recommended by humble and Mozelius (2022) for an understanding that is very thorough through analysis of verbatim transcripts so that the understanding can be the ground. By employing this strategy, the writer can identify major ideas, which is useful for grasping the issue clearly. The interview responses were transcribed verbatim and then systematically coded and classified to enable comprehensive and intensive analysis (Walle, 2015).

The thematic analysis followed the six-stage process outlined by Braun and Clarke (2022). The six-step process involved: familiarisation with the data, generating initial codes, searching for themes, reviewing themes, defining and naming themes, and producing the report. All the data was carefully examined and coded, and related concepts were grouped into key themes that emerged from the data (Wakelin, McAra-Couper and Fleming, 2024). Transcripts from online interviews with 6 participants and responses to questionnaires from 2 participants were analysed. All these participants were involved in various sector to provide a generalized opinion on the impact of the policies and the diverse perspective of the individuals. This analysis identified major themes, such as government support and policy implementation. The data shed light on the research objectives, which may guide decision-making for policymakers and provide strategic guidance for entrepreneurs during times of crisis. In addition to the interview analysis, the second part of the study involved examining key policies such as Enterprise Ireland and Startup India, which were designed to promote the entrepreneurial ecosystem in their respective countries (Henry, Hill and Leitch, 2003).

4.6. Ethical Considerations

Ethics is the code of morals that defines how people should act and behave in society or the conduct of an activity. It is important to maintain integrity during the conduct of research, protecting the rights and welfare of participants, and also upholding research credibility are very important elements which should be observed while carrying out any research ethically (Wienclaw, 2021). One of the major ethical prerequisites in research is obtaining informed consent from the participants, because it shows an appreciation for their autonomy and ensures they know why this study is being carried out and what its procedures are, before deciding on whether or not to take part. All the interviews were conducted adhered to the prescribed Ethical Guidelines and Procedures for Research involving Human Participants as set out by National College of Ireland. Participants were made aware of what the research was about and why it was being done for a master's degree, and they agreed to have their conversations recorded. In accordance with ethical guidelines and procedures, all recordings and transcripts will be kept in a folder protected by passwords for 5 years before deletion. The ethical form from NCI which included details of research procedures was completed properly; it was submitted together with the proposal on this study.

5. FINDINGS & ANALYSIS

This study examined government initiatives in India and Ireland designed to support entrepreneurs during the COVID-19 pandemic. The findings reveal both commonalities and differences in policy approaches and their effectiveness. The study, with its holistic approach to the research objective as outlined in Section 4, provides insights into the aims of the government policies and initiatives. The detailed content analysis indicates that the responses from the two governments were largely similar, though the effects of their policies differed to some extent.

5.1. Difficulties in accessing the benefits of policy initiatives

The study reveals significant challenges entrepreneurs face in accessing the benefits of government policy initiatives designed to support and promote business growth. Both India and Ireland have launched various schemes, such as Startup India and Enterprise Ireland, aimed at fostering new businesses and aiding entrepreneurs during the pandemic. However, despite awareness of these initiatives, many participants encountered substantial obstacles in effectively leveraging these benefits. The studies by Fikri and Newman (2024) and Chaturvedi and Goyal (2012) highlight factors such as bureaucratic hurdles, complex application processes, and a lack of transparency in eligibility criteria as key impediments that undermine the accessibility and impact of such programs.

Participant 2, Founder of EVA Shipping, an entrepreneur from India, expressed deep frustration with the challenges of accessing government grants, despite meeting all the eligibility criteria. Her experience serves to highlight a significant issue where bureaucratic complexities and administrative hurdles can prevent eligible entrepreneurs from obtaining the crucial financial support they desperately need to sustain and grow their businesses, particularly during times of crisis or economic uncertainty.

"I have registered with the Startup India programme run by the central government and am waiting to receive their loan. However, I am unsure how positive this process will be. I have been told that if you hold a Startup India certificate, you can approach banks who are

then eligible to provide you with an 80% loan based on your business plan or idea. Unfortunately, so far, I have approached several banks but none of them seem to be aware of the Startup India scheme." - Participant 2, Founder of EVA Shipping

Similarly, Participant 6, Founder of Eccospray, from Ireland commended the loan initiatives available to businesses but pointed out the considerable difficulties in securing funds from banks. This suggests that while financial support mechanisms may be in place, practical challenges in the application and approval processes can impede access, undermining the effectiveness of these initiatives.

"The government's policy fails to recognise that those who have learned from their failures may have a better chance of succeeding in their next venture. However, without access to funding through traditional banks, there are alternative options, such as the Microfinance Bank of Southern Ireland, which can provide funding to those who have been refused credit by mainstream banks." - Participant 6, Founder of Eccospray

On the other hand, Participant 4, Founder of RA Energy, who is involved in renewable energy projects, had a more positive experience. His solar project benefited from extensive grants provided by the Government of India, with additional funding facilitated through a collaboration between the Bank of India and the World Bank. This example illustrates that while some sectors, such as renewable energy, might find more streamlined access to support due to aligned policies and targeted funding efforts, this level of access may not be uniformly experienced across all sectors.

The finance ministry played a critical role in supporting the renewable energy sector in 2019. The Government of India, the State Bank of India, and the World Bank made a mutual agreement to fund the solar business in India. - Participant 4, Founder of RA Energy

Simultaneously, Participant 8, Founder of Oracawise AI, from Ireland is aware of the government grant schemes but did not avail of the benefits. Similarly, Participant 5,

Founder of Ark Consultants, from India who is involved in aerospace and defence reported that military spending was drastically reduced by almost all countries, with funds and facilities diverted towards healthcare and saving lives. The only policy change was in terms of support towards leeway in payment timelines and delivery schedule/period extensions. In both cases, despite not availing the government supports or grants, they were able to navigate the challenges without much friction.

Overall, the difficulties in accessing these policy benefits reveal a need for improvements in the implementation and administration of support schemes. Streamlining processes, enhancing transparency, and addressing specific sectoral challenges could help ensure that government initiatives more effectively reach and support the entrepreneurs they are designed to assist.

5.2. Financial support is essential to survive the business at the time of crisis:

A key finding of this study is the essential role of financial support in helping businesses survive during times of crisis. The pandemic placed unprecedented strain on businesses, which were at various stages of development when the crisis hit (Das and Rangarajan, 2020). Government policies and initiatives, including tax deferrals and financial incentives, proved crucial in preventing widespread business closures across both India and Ireland.

For instance, Participant 7. Founder Connect Promotions, from Ireland shared how the government's incentives and allowances played a critical role in sustaining his business during the pandemic. The financial support enabled him to continue paying his employees' salaries, which was instrumental in maintaining operational stability and navigating the economic downturn. This support not only helped Participant 7's business survive but also preserved jobs and provided a buffer against the severe economic impacts of the crisis.

"The second scheme was the EWSS i.e. Employment Wage Subsidy Scheme, which provided payments to workers who had lost earnings. We utilised this to help maintain people's incomes. This was a definite help, and I think it would have been much more difficult without this support." - Participant 7, Founder of Connect Promotions

Similarly, the study reveals that across the participants in both countries benefited from various forms of financial assistance. These measures were designed to offer immediate relief and stabilize businesses by alleviating some of the financial pressures brought on by the pandemic. The effectiveness of these initiatives highlights the importance of timely and targeted financial interventions in safeguarding the viability of businesses during economic disruptions.

During the COVID-19 pandemic, entrepreneurs in India and Ireland were able to benefit from temporary tax relief, where they were only required to pay mandatory taxes for a limited time. This financial support encouraged them to keep their businesses operational and facilitated the revitalisation and revamping of their operations once the situation returned to normal (Jabłońska *et al.*, 2021).

"We get lot of benefits including the No GST from here. OK, so that's a big saving for us and you know from time to time probably government can do a lot in in doing such activities. So that all the people who are creating jobs there should be incentivized more." - Participant 1, Founder of Orhai

In summary, the study emphasizes that government financial support was not merely beneficial but essential for business continuity during the crisis. The ability to access such support played a decisive role in whether businesses could weather the storm, underscoring the critical need for robust and responsive financial assistance mechanisms in times of crisis.

5.3. Entrepreneurial attitude is equally important to overcome the crisis

A key finding of this study is the vital role that an entrepreneurial attitude plays in navigating and overcoming crises. Beyond the direct support provided by government policies and initiatives, the mindset and adaptability of entrepreneurs are crucial for business resilience and recovery.

For instance, Participant 3, Founder of Skill Live, from India, experienced significant challenges when her company, which specialised in teaching the Chinese language,

faced a strong public backlash linked to the pandemic's origins. This backlash was exacerbated by the deteriorating political relations between India and China, further impacting her business. Faced with these adversities, Participant 3demonstrated remarkable agility by swiftly restructuring her company to better align with the changing circumstances. She was able to transform her business in a short period of time by strategically changing the business model from teaching Chinese to teaching other foreign languages, such as Spanish and French. Her experience highlights how an entrepreneurial attitude which is characterised by resilience, flexibility, and proactive problem-solving, can be instrumental in navigating unexpected crises.

In early 2020, we learned about the COVID-19 pandemic. I'll share a personal experience regarding my company, without mentioning anyone else. The company's name was YehChina, chosen to allow people to experience Chinese culture and language. The word "Yeh" in Hindi means "this", so it was intended to create a miniature China experience. However, the COVID-19 outbreak greatly impacted us due to the company name. We faced significant aggression from people and customers, who questioned the "China" part of the name because of the pandemic's association with the country. This was a challenging situation that we had to navigate. - Participant 3, Founder of Skill Live

Similarly, Participant 1 from India emphasizes that while government support is essential, the role of the entrepreneur's attitude cannot be underestimated. In times of crisis, it is ultimately the entrepreneur who determines how effectively to lead and adapt their business. This underscores that government initiatives, while important, are only part of the equation. The entrepreneurial mindset which encompassing qualities such as innovation, determination, and the willingness to pivot will profoundly influences a business's ability to survive and thrive during challenging times (Kolvereid and Isaksen, 2012).

In short, while government policies provide necessary financial and structural support, the entrepreneurial attitude significantly impacts how effectively businesses can utilize these resources and adapt to the evolving landscape during a crisis.

5.4. Limitations of the Findings:

This study's findings are based on subjective perspectives and experiences of the participants. Though qualitative analysis was employed, it would have been more advantageous to use both qualitative and quantitative methods. However, due to time constraints, the researchers were limited to qualitative content analysis alone. Despite addressing the problems faced and the countermeasures adopted, the response to the crisis varies across different sectors, and the market situation and demand also play a significant role, which was not covered in this study.

6. DISCUSSION

The COVID-19 pandemic has presented an unexpected set of challenges for entrepreneurial ecosystems globally, testing the capability of any government to sustain businesses in such a situation. This study, is an attempt to explore how effective such interventions are, this research combines literature review with an examination of various initiatives launched by the Indian and Irish governments with key inputs from entrepreneurs in both nations. By examining the nuances of policy implementation and the varying experiences of entrepreneurs in these diverse contexts, we uncover a complex landscape of support mechanisms and their impacts.

6.1. Challenges in Accessing Government Benefits

Both India and Ireland had full policy frameworks that would cushion entrepreneurs against the socio-economic adverse effects of the pandemic. The availability of these benefits is a challenge identified by this study. It is an impact brought about by operational challenges in bureaucratic and administrative procedures faced by entrepreneurs in both countries, hindering the full utilization of available support.

In India, the "Startup India" and "Make in India"-type initiatives remained entrapped in red tape, along with a general lack of awareness about these initiatives among the stakeholders. For example, Participant 2, founder of EVA Shipping from India, shared her experience, which underlines how long-winded administrative processes and fragmented communication between government bodies and financial institutions cause delays and frustrations. Similarly, in Ireland, Participant 6's observations highlight that despite the availability of support mechanisms like the Employment Wage Subsidy Scheme, the practicalities of accessing these benefits were marred by inefficiencies and a lack of coordination.

This observation is consistent with findings by Fikri and Newman (2024), that complex administrative procedures and lack of proper communication channels may strongly affect the effectiveness of policy interventions. In view of the foregoing challenges, the governments have to heighten transparency, fast-track application processes, and ensure good coordination between the different entities involved in

implementing these policies. This would improve the aspects whereby support would reach the beneficiaries more efficiently, hence amplifying the impact of the policies.

6.2. The Essential Role of Financial Support

The study, therefore, confirms the crucial role of financial support in business survival during a crisis. Financial aids instituted in both India and Ireland included tax deferrals, subsidies, and grants, which were quite instrumental in helping businesses stay afloat and thus saving jobs. For example, the experience of Participant 7 with the Wage Subsidy Scheme in Ireland and the insights of Participant 1 into the tax-relief measures taken by India show how these financial supports were instrumental in riding out the economic challenges posed by the pandemic.

Research supports the notion that timely and targeted financial interventions are crucial during periods of economic disruption. Such financial assistance gives businesses a chance to survive shocks and continue operations (Das and Rangarajan, 2020; Demirel and Danisman, 2019). External financial support acts as a buffer, mitigating the adverse effects of economic downturns and helping preserve both employment and business continuity. This highlights the necessity of robust financial mechanisms as a core component of any crisis management strategy.

6.3. Sector-Specific Variations in Policy Effectiveness

The study also highlights that the effectiveness of government support varied significantly across different sectors. While Participant 4's positive experience with renewable energy grants in India demonstrates how targeted support can benefit specific industries, other sectors struggled with a lack of alignment or inadequate support. This sector-specific variation underscores the need for a more nuanced approach to policy design.

Different industries have got unique requirements and challenges that should be taken into consideration in effective policy interventions. When the governments customize their support mechanisms to serve the specific needs of various sectors, they can make their policy overall impacts greater. In addition, making available such sector-specific guidance and resources ensures that all areas of the economy are well-equipped to managing crises hence facilitating a more resilient economic atmosphere.

6.4. Comparative Analysis of Policy Approaches

The comparative analysis of policy responses between India and Ireland reveals differing strategies in fostering entrepreneurship. Ireland focused on reducing regulatory burdens and providing financial incentives, while India emphasized digital empowerment and infrastructure development. These contrasting approaches reflect the unique economic landscapes and developmental priorities of each country. The importance of minimizing bureaucratic hurdles and providing accessible financial assistance can be seen in Ireland's approach. Ireland has proven that reducing the administrative burden can improve the effectiveness of help by streamlining regulatory processes and offering specific financial aid. Conversely, through "Startup has TN", India emphasizes the possibility of using technology and local initiatives as a way to develop entrepreneurship through digital infrastructure. It has been observed in this comparative perspective that a balanced policy framework consisting of both approaches will create an all-encompassing support system for entrepreneurs. Combining efficient administrative processes with robust digital and infrastructural support can create a more resilient and responsive entrepreneurial ecosystem.

6.5. Implications for Future Policy Formulation

The main findings of the study have huge implications for future policy aimed at assisting entrepreneurship. In general, policy needs to be developed under more emphasis in the following areas by governments if it is to be more effective:

1. Administrative Streamlining Administrative Process: The application procedures should be simplified and more transparent; it would reduce bureaucratic delays and offer broader access to support. Improved communication among government agencies and financial institutions to ensure the effective implementation of support mechanisms is also essential.

- 2. Improving Sector-Based Support: By designing policies that take into account the unique requirements of different industries, their effectiveness can be improved. Specific guidance and funding for sectors can help to meet the varied challenges they face, resulting in more precise and meaningful interventions.
- 3. Public-Private Collaboration: Public-private collaboration can play a very important role in making these support programs for entrepreneurs effective. The different successful experiences in India and Ireland have shown enough evidence that capitalizing on resources and expertise from the different stakeholders will definitely put up a more robust support system.
- **4.** Adaptability and Flexibility: Policies should be able to take into account changed circumstances and evolving needs. This shall include flexibility in the financial support mechanisms so that new challenges and opportunities can be faced appropriately upon their emergence.

In conclusion, it can be said that both the strengths and weaknesses of governmental support systems for entrepreneurship have been emphasized by the COVID-19 pandemic. Both countries, India as well as Ireland introduced different measures for businesses during artificial impending crises, however; they proved to be less effective due to bureaucratic bottlenecks, challenges specific to various sectors and policy implementation variations among nations. In addressing these challenges and enhancing the current effective areas in policies, governments will be able to foster better entrepreneurial ecosystems. It not only helps when dealing with emergencies but also promotes long-term growth and innovation in economies.

The lessons learned from India's and Ireland's experiences may now guide future policy making efforts that intend to more effectively support entrepreneurs in order to create resilient and dynamic economies. To conclude it's important to note that entrepreneurship promotion requires a comprehensive approach during tough times transcending simple designs of policies themselves. These include efficient execution, customized support systems coupled with an emphasis on reducing red tape so as to make an enabling environment for sustainability within economies through entrepreneurship. By learning from both successes and shortcomings,

policymakers can build more resilient and adaptive support systems capable of withstanding future challenges and driving innovation.

7. CONCLUSION & RECOMMENDATIONS

This comparative analysis of Irish and Indian entrepreneurial ecosystems gives emphasis on how some governments intervene in entrepreneurship during times of economic turmoil. Both countries had a lot of different government programs aimed at helping entrepreneurs but they took different approaches in promoting these policies. Again, this shows that the effectiveness of programs is more important than their formulation for they must be directed to those who need them the most.

The study findings emphasize that multi-dimensional approach in entrepreneurship ecosystem support includes financial assistance, simplified laws, infrastructure development and a culture of innovation. However, it is not enough just to have it's policies. For translating policy into real life benefits, effective implementation usually needing inter-agency coordination, simplified processes and targeted outreach is crucial. This particular study is the foundation for many future research endeavours concerning entrepreneurial ecosystems, with an interesting way being directed to sector-specific lenses. By focusing on specific sectors, such as fast-growing sectors like the renewable energy, future researchers would assist researchers to provide extensive and practical proposals for policy makers. This would yield much better and more effective interventions that promote innovation, economic growth, and enhance environmental sustainability.

The comparison of India and Ireland reveals how government policies shape the entrepreneurial landscape amid economic crises. Furthermore, this research also shows that there is need for global cooperation and adaptation of policies from the best practices so as to improve on different entrepreneurial systems within varying socio-economic environments. Through an exploration of the experiences of these two nations, this research provides useful information about how public policies can create, develop and sustain entrepreneurial mindsets during periods of economic distress with pragmatic suggestions for directing officers on both sides and further. In conclusion, the research presents an all-inclusive comprehension of how government policy influences entrepreneurship cultivation by showing workings amid whose effect are enacted policy choice, economic robustness and entrepreneur success during global economic challenges.

References

Abdelkafi, R. and Refas, S. (2021) 'Out with the old or out with the new: The uncertain role of entrepreneurship in developing countries in the post-COVID-19 context', *Theoretical Economics Letters*, 11(5), pp. 1002-1019. doi: 10.4236/tel.2021.115064.

Abrams, R. M. (2007) 'Prophet of innovation: Joseph Schumpeter and creative destruction by Thomas K. McCraw', *Business History Review*, 81(4). Available at: https://research.ebsco.com/linkprocessor/plink?id=f92db62f-b6e6-30a2-9514-92b65d3aa252 [Accessed 25 July 2024].

Abutabenjeh, S. and Jaradat, R. (2018) 'Clarification of research design, research methods, and research methodology: A guide for public administration researchers and practitioners', *Teaching Public Administration*, 36(3), pp. 237-258. doi: 10.1177/0144739418775787

AIB (2024) SBCI Future Growth Loan Scheme. Available at: https://aib.ie/business/sbci/sbci-future-growth-loan-scheme#contact [Accessed 25 July 2024].

Alvarez, S. A. and Barney, J. B. (2014) 'Entrepreneurial opportunities and poverty alleviation', *Entrepreneurship: Theory & Practice*, 38(1), pp. 159-184. doi:10.1111/etap.12078.

Alvarez, S. A. and Busenitz, L. W. (2001) 'The entrepreneurship of resource-based theory', *Journal of Management*, 27(6), pp. 755-775. doi: 10.1177/014920630102700609

Audretsch, D. B., Keilbach, M. C. and Lehmann, E. E. (2006) *Entrepreneurship and economic growth*. Oxford: Oxford University Press.

Benlaria, H., Almawishir, N. F. S., Saadaoui, S., Mohammed, S. M. M., Abdulrahman, B. M. A. and Eltahir, I. A. E. (2023) 'The moderating role of research and development (R&D) support in the relationship between entrepreneurship and per capita output—A study on the GCC countries', *Economies*, 11(6), p. 162. doi:10.3390/economies11060162.

Blanchflower, D. G. and Oswald, A. J. (1998) 'What makes an entrepreneur?', *Journal of Labor Economics*, 16(1), pp. 26-60. doi:10.1086/209881.

Boudreaux, C. J., Jha, A. and Escaleras, M. (2023) 'Natural disasters, entrepreneurship activity, and the moderating role of country governance', *Small Business Economics*, 60(4), pp. 1483–1508. doi:10.1007/s11187-022-00657-y.

Brattström, A. and Wennberg, K. (2022) 'The entrepreneurial story and its implications for research', *Entrepreneurship: Theory & Practice*, 46(6), pp. 1443–1468. doi:10.1177/10422587211053802.

Braun, V. and Clarke, V. (2022) *Thematic analysis: A practical guide*. London: Sage Publications Ltd. Available at: https://research.ebsco.com/linkprocessor/plink?id=10b44daa-c24e-3484-ad6d-7eb83eb6c6ec [Accessed 29 July 2024]

Bullard, E. (2023) 'Purposive sampling', *Salem Press Encyclopedia*. Available at: https://research.ebsco.com/linkprocessor/plink?id=ed9cc795-6464-3c13-946c-1c197f768ee1 [Accessed 29 July 2024).

Cadar, O. and Badulescu, D. (2015) 'Entrepreneur, entrepreneurship and intrapreneurship. A literature review', *Annals of the University of Oradea, Economic Science Series*, 2(24), pp. 658-664. Available at: https://research.ebsco.com/linkprocessor/plink?id=8b3c8705-6a3f-388e-8106-09e2c10bcc74 [Accessed 25 June 2024].

Chaturvedi, A. and Goyal, S. (2012) 'Changing severity of impediments for entrepreneurs in India: an empirical analysis', *International Journal of Entrepreneurship and Small Business*, 16(1), pp.1-32. doi: 10.1504/IJESB.2012.046914

Das, M. and Rangarajan, K. (2020) 'Impact of policy initiatives and collaborative synergy on sustainability and business growth of Indian SMEs', *Indian Growth and Development Review*, 13(3), pp. 607-627. doi:10.1108/IGDR-09-2019-0095.

Demirel, P. and Danisman, G. O. (2019) 'Eco-innovation and firm growth in the circular economy: Evidence from European small- and medium-sized enterprises', *Business Strategy & the Environment*, 28(8), pp. 1608–1618. doi:10.1002/bse.2336.

Department of Social Protection (2023) *Operational guidelines: Enterprise Support Grant*. Available at: https://www.gov.ie/en/publication/80a39-operational-guidelines-enterprise-support-grant/#12-objective [Accessed 12 July 2024].

Dingliwal, U. and Sharma, S. (2023) 'The impact of Covid-19 pandemic on startups in India', *Bayan College International Journal of Multidisciplinary Research*, 3(2), pp. 94-108. Available at: https://www.researchgate.net/publication/371761124 [Accessed 15 January 2024].

Engel, J. S. (2015) 'Global clusters of innovation: Lessons from Silicon Valley', *California Management Review*, 57(2), pp. 36-65. doi:10.1525/cmr.2015.57.2.36.

Enterprise Ireland (2024) *Sustaining Enterprise Fund*. Available at: https://globalambition.ie/supports/innovation-support/sustaining-enterprise-fund/ [Accessed 28 July 2024].

Fikri, K and Newman, D. (2024) *How the pandemic rebooted entrepreneurship in the U.S.* Available at: https://hbr.org/2024/01/how-the-pandemic-rebooted-entrepreneurship-in-the-u-s [Accessed 01 July 2024].

Gan, T., Zhang, M. and Zhang, Z. (2023) 'The impact of digital government policy on entrepreneurial activity in China', *Economic Analysis and Policy*, 79, pp. 479-496. doi:10.1016/j.eap.2023.06.029.

Gashenko, I. V. V., Vokina, S. G., Romanov, D. G., Bezrukova, T. L. and Kozenko, Y. A. (2016) 'Theoretical and methodological aspects of innovation development in modern economic systems', *Contemporary Economics*, 10(4), pp. 363-372. doi:10.5709/ce.1897-9254.223.

Gigauri, I. and Damenia, N. (2020) 'Cooperation between social entrepreneurs and government to develop solutions to social problems', *Business and Economic Research*, 10(3), pp. 116-136. doi:10.5296/ber.v10i3.17383

Goswami, N., Murti, A. B. and Dwivedi, R. (2023) 'Why do Indian startups fail? A narrative analysis of key business stakeholders', *Indian Growth and Development Review*, 16(2), pp. 141–157. doi:10.1108/IGDR-11-2022-0136.

Gouvea, R., Kapelianis, D. and Montoya, M. (2018) 'Marketing challenges and opportunities in emerging economies: A Brazilian perspective', *Thunderbird International Business Review*, 60(2), pp. 193-205. doi:10.1002/tie.21840.

Grant, R. M. (1991) 'The resource-based theory of competitive advantage: Implications for strategy formulation', *California Management Review*, 33(3), pp. 114-135. doi:10.2307/41166664.

Hamdan, A. M. M. (2019) 'Entrepreneurship and economic growth: An Emirati perspective', *Journal of Developing Areas*, 53(1), pp. 65-78. doi:10.1353/jda.2019.0004.

Henry, C., Hill, F. and Leitch, C. (2003) 'Developing a coherent enterprise support policy: A new challenge for governments', *Environment and Planning C: Government and Policy*, 21(1), pp. 3-19. doi: 10.1068/c0220

Humble, N. and Mozelius, P. (2022) 'Content analysis or thematic analysis: Doctoral students' perceptions of similarities and differences', *Electronic Journal of Business Research Methods*, 20(3), pp. 89–98. doi:10.34190/ejbrm.20.3.2920.

Ingram, N. (2018) 'Tesla Motors: A potentially disruptive force in a mature industry', *International Journal of Management and Applied Research*, 5(1), pp. 8-22. doi: 10.18646/2056.51.18-002

Jabłońska, M., Stawska, J., Dziuba, R., Tekce, M. and Krasoń, M. (2021) 'Public aid and entrepreneurship during the Covid-19 pandemic in the European Union countries', *Finanse i Prawo Finansowe*, 3(31), pp. 57-77.

Jahanshahi, A. A., Nawaser, K., Khaksar, S. M. S. and Kamalian, A. R. (2011) 'The relationship between government policy and the growth of entrepreneurship in the micro, small & medium enterprises of India', *Journal of Technology Management & Innovation*, 6(1), pp. 66-76. doi:10.4067/S0718-27242011000100007.

Javadian, G., Dobratz, C., Gupta, A., Gupta, V. K. and Martin, J. A. (2020) 'Qualitative research in entrepreneurship studies: A state-of-science', *The Journal of Entrepreneurship*, 29(2), pp. 223-258. doi: 10.1177/0971355720930564

Khyareh, M. M. and Amini, H. (2021) 'Governance quality, entrepreneurship and economic growth', *Journal of Competitiveness*, 13(2), pp. 41-64. doi:10.7441/joc.2021.02.03.

Koch, J. L. and Caradonna, T. M. (2006) 'Technologies and business models that work in developing countries', in *2006 International Conference on Information and Communication Technologies and Development*. Berkeley, CA, USA, 25-26 May 2006, pp. 193-202. doi:10.1109/ICTD.2006.301857.

Kolvereid, L. and Isaksen, E. J. (2012) 'The psychology of the entrepreneur', in Mole K. and Ram, A. (eds.) *Perspectives in entrepreneurship: A critical approach*. London: Red Globe Press, pp. 27-39.

Koster, S. and Rai, S. K. (2008) 'Entrepreneurship and economic development in a developing country: A case study of India', *The Journal of Entrepreneurship*, 17(2), pp. 117-137. doi: 10.1177/097135570801700202

Kratochwil, F. (2008) 'Constructivism: What it is (not) and how it matters', in Della Porta, D. and Keating, M. (eds.) *Approaches and methodologies in the social sciences–A pluralist perspective*. Cambridge: Cambridge University Press, pp. 80-98.

Krishnan, C.S.N., Ganesh, L.S. and Rajendran, C. (2022) 'Entrepreneurial Interventions for crisis management: Lessons from the Covid-19 Pandemic's impact on entrepreneurial ventures', *International Journal of Disaster Risk Reduction*, 72. doi:10.1016/j.ijdrr.2022.102830.

Kuckertz, A., Brändle, L., Gaudig, A., Hinderer, S., Reyes, C. A. M., Prochotta, A., Steinbrink, K. M. and Berger, E. S. (2020) 'Startups in times of crisis – A rapid response to the COVID-19 pandemic', *Journal of Business Venturing Insights*, 13, e00169. doi:10.1016/j.jbvi.2020.e00169.

Kumar, S., Maheshwari, V., Prabhu, J., Prasanna, M., Jayalakshmi, P., Suganya, P., Malar, B. A. and Jothikumar, R. (2020) 'Social economic impact of COVID-19 outbreak in

India', International Journal of Pervasive Computing and Communications, 16(4), pp. 309-319. doi:10.1108/IJPCC-06-2020-0053.

Laing, E., van Stel, A. and Storey, D. J. (2022) 'Formal and informal entrepreneurship: A cross-country policy perspective', *Small Business Economics*, 59(3), pp. 807-826. doi:10.1007/s11187-021-00548-8.

Lapadat, J. C. and Lindsay, A. C. (1999) 'Transcription in research and practice: From standardization of technique to interpretive positionings', *Qualitative Inquiry*, 5(1), pp. 64-86. doi:10.1177/107780049900500104.

Lee, H. J. (2019) 'What factors are necessary for sustaining entrepreneurship?', *Sustainability (Switzerland)*, 11(11), 3022. doi:10.3390/su11113022.

Lima, J. C. F., Torkomian, A. L. V., Pereira, S. C. F., Oprime, P. C. and Hashiba, L. H. (2021) 'Socioeconomic impacts of university-industry collaborations-A systematic review and conceptual model', *Journal of Open Innovation: Technology, Market, and Complexity*, 7(2), 137. doi:10.3390/joitmc7020137.

Lin, G. T. R., Shen, Y.-C. and Chou, J. (2010) 'National innovation policy and performance: Comparing the small island countries of Taiwan and Ireland', *Technology in Society*, 32(2), pp. 161-172. doi:10.1016/j.techsoc.2010.03.005.

Liu, A. M., Liang, O. X., Tuuli, M. and Chan, I. (2018) 'Role of government funding in fostering collaboration between knowledge-based organizations: Evidence from the solar PV industry in China', *Energy Exploration & Exploitation*, 36(3), pp. 509-534. doi: 10.1177/0144598717742968

Low, L. (2005) 'Entrepreneurship development in Ireland and Singapore', *Journal of the Asia Pacific Economy*, 10(1), pp. 116–138. doi:10.1080/1354786042000309107.

Lowrey, Y. (2003) 'The entrepreneur and entrepreneurship: A neoclassical approach', Office of Advocacy, US Small Business Administration Economic Research Working Paper. Available at: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=744785 [Accessed 7 August 2024].

Maguire, M. and Delahunt, B. (2017) 'Doing a thematic analysis: A practical, step-by-step guide for learning and teaching scholars', *AISHE-J: The All Ireland Journal of Teaching & Learning in Higher Education*, 9(3), pp. 3351-33514. Available at: https://research.ebsco.com/linkprocessor/plink?id=ce3141fd-4693-398f-9f01-6eb6a027d111 [Accessed 29 July 2024].

Martínez, J. A. B. (2009) 'Entrepreneurship policies and new business performance and growth: An empirical analysis based on manufacturing industries', *Environment and Planning C: Government and Policy*, 27(2), pp.195-215. doi: 10.1068/c0816b

Matenda, F. R. and Sibanda, M. (2023) 'The influence of entrepreneurship on economic growth in BRICS economies', *Economic Research-Ekonomska Istrazivanja*, 36(3), pp. 1-21. doi:10.1080/1331677X.2023.2275582.

McClure, J. E. and Thomas, D. C. (2021) 'Schumpeter's fatalistic view of capitalism: How his "essential" process of creative destruction survived', *Journal of Private Enterprise*, 36(2), pp. 77-102. Available at: https://research.ebsco.com/linkprocessor/plink?id=e015c7a9-c8d8-34da-ac17-5aee9b36208f [Accessed 25 January 2024].

Meenakshi, N. and Sinha, A. (2019) 'Food delivery apps in India: Wherein lies the success strategy?', *Strategic Direction*, 35(7), pp. 12-15. doi:10.1108/SD-10-2018-0197.

Mehra, A., Mathur, N. and Tripathi, V. (2019) 'Sahaj Crafts: The challenge of alleviating poverty in Western Rajasthan', *Emerald Emerging Markets Case Studies*, 9(1), pp. 1-45. doi:10.1108/EEMCS-06-2018-0099.

Mishra, A. (2024) 'PM MUDRA Yojana: How to get a collateral-free loan with this Scheme', *Business Standard*, 10 April. Available at: https://www.business-standard.com/finance/personal-finance/pm-mudra-yojana-how-to-get-a-collateral-free-loan-with-this-scheme-124041000312_1.html [Accessed 25 July 2024].

Misra, R., Mahajan, R., Singh, N., Khorana, S. and Rana, N. P. (2022) 'Factors impacting behavioural intentions to adopt the electronic marketplace: Findings from small businesses in India', *Electronic Markets*, 32(3), pp. 1639-1660. doi:10.1007/s12525-022-00578-4.

Muegge, S. and Reid, E. (2019) 'Elon Musk and SpaceX: A case study of entrepreneuring as emancipation', *Technology Innovation Management Review*, 9(8), pp. 18-29. doi:10.22215/timreview/1258.

Nayak, Y. D and Sahoo, A. P. (2021) 'Green entrepreneurship in India', *International Journal of Agricultural Research, Innovation and Technology*, 11(2), pp. 72-75. doi:10.3329/ijarit.v11i2.57258.

Neumeyer, X., Ashton, W. S. and Dentchev, N. (2020) 'Addressing resource and waste management challenges imposed by COVID-19: An entrepreneurship perspective', *Resources, Conservation & Recycling*, 162, 105058. doi:10.1016/j.resconrec.2020.105058.

Newman, A., Obschonka, M. and Block, J. (2022) 'Small businesses and entrepreneurship in times of crises: The renaissance of entrepreneur-focused micro perspectives', *International Small Business Journal: Researching Entrepreneurship*, 40(2), pp. 119-129. doi: 10.1177/02662426211063390.

Nuthalapati, C. S. and Nuthalapati, C. (2021) 'Has open innovation taken root in India? Evidence from startups working in food value chains', *Circular Economy and Sustainability*, 1(4), pp. 1207-1230. doi: 10.1007/s43615-021-00074-5

OECD (2021) 'Tax and fiscal policies after the COVID-19 crisis', OECD Policy Responses to Coronavirus (COVID-19). doi: 10.1787/5a8f24c3-en.

Pandey, I. M., Nair, R., Awasthi, D., Mehta, K., Varshney, V., Rewari, R. and Ramachandran, K. (2003) 'Entrepreneurship and venture capital', *Vikalpa*, 28(1), pp. 99-112. doi: 10.1177/0256090920030109

Pickard, A. J. (2013) *Research methods in information*. 2nd edn. London: Facet Publishing.

Polyak, P. (2022) 'Ireland's multinationals-dominated economy in the pandemic: Did big tech and big pharma save the day?', *International Journal of Political Economy*, 51(1), pp. 65-76. doi:10.1080/08911916.2022.2046347.

Roller, M. R. and Lavrakas, P. J. (2015) *Applied qualitative research design: A total quality framework approach*. EBSCO eBook Collection. Available at: https://research.ebsco.com/linkprocessor/plink?id=f4b44073-a763-3364-a365-087dd373c408 [Accessed 29 July 2024].

Santoso, R. and Fianto, A. Y. A. (2022) 'Creative industry and economic recovery strategies from pandemic disruption', *Jurnal Ilmu Ekonomi Terapan*, 7(1), pp. 47-62. doi:10.20473/jiet.v7i1.35008.

Saunders, M. N. K., Thornhill, A. and Lewis, P. (2019) *Research methods for business students*. 8th edn. VLeBooks. Available at: https://research.ebsco.com/linkprocessor/plink?id=afe0f395-56d3-3b7f-b863-e73a7cc67dd7 [Accessed 15 June 2024].

Shafiu, A. M., Manaf, H. A. and Muslim, S. (2020) 'Utilization entrepreneurship for job creation, poverty reduction and national development', *The Journal of Social Sciences Research*, 6(1), pp. 97-102. doi: 10.32861/jssr.61.97.102

Shkabatur, J., Bar-El, R. and Schwartz, D. (2022) 'Innovation and entrepreneurship for sustainable development: Lessons from Ethiopia', *Progress in Planning*, 160, 100599. doi:10.1016/j.progress.2021.100599.

Sindakis, S. and Showkat, G. (2024) 'The digital revolution in India: Bridging the gap in rural technology adoption', *Journal of Innovation & Entrepreneurship*, 13(1), 29. doi:10.1186/s13731-024-00380-w.

Singh, A. S., Venkataramani, B. and Ambarkhane, D. (2021) Post-pandemic penury of the financially marginalized in India: Coping with the new normal, in *The future of*

service post-COVID-19 pandemic, Volume 1, pp. 201-225. doi:10.1007/978-981-33-4126-5_10.

Startup India (2024) What is the Startup India initiative? Available at: https://www.startupindia.gov.in/content/sih/en/about-startup-india-initiative.html [Accessed 30 June 2024].

Tandon, A., Gupta, A., Goel, P. and Singh, V. K. (2020) 'Impact of digitisation on entrepreneurial ecosystems: an Indian perspective', *International Journal of Business and Globalisation*, 25(2), pp. 154-169. doi: 10.1504/IJBG.2020.107887

Tuszynski, M. P. and Stansel, D. (2018) 'Targeted state economic development incentives and entrepreneurship', *Journal of Entrepreneurship and Public Policy*, 7(3), pp. 235–247. doi:10.1108/JEPP-D-18-00033.

Vadeboncoeur, L. R. and Jennifer, A. (2023) 'Institutional theory', *Salem Press Encyclopedia*. Available at: https://research.ebsco.com/linkprocessor/plink?id=32bc3174-a8c4-3a79-9bba-4497b088a13f [Accessed 29 July 2024].

Vatavu, S., Dogaru, M., Moldovan, N.-C. and Lobont, O.-R. (2022) 'The impact of entrepreneurship on economic development through government policies and citizens' attitudes', *Economic Research-Ekonomska Istrazivanja*, 35(1), pp. 1604–1617. doi: 10.1080/1331677X.2021.1985566.

Wakelin, K. J., McAra-Couper, J. and Fleming, T. (2024) 'Using an online platform for conducting face-to-face interviews', *International Journal of Qualitative Methods*, pp. 1–9. doi:10.1177/16094069241234183.

Walle, A. H. (2015) *Qualitative research in business: A practical overview*. EBSCO eBook Collection. Available at: https://research.ebsco.com/linkprocessor/plink?id=0bd91315-2993-300b-91fd-34b224182d7b [Accessed 29 July 2024].

Wienclaw, R. A. (2021) 'Research ethics in sociology', *Salem Press Encyclopedia*. Available at: https://research.ebsco.com/linkprocessor/plink?id=c3dfc70f-4358-3904-baed-23133c86f30e [Accessed 29 July 2024].

Williams, D. (2013) 'Building a support infrastructure for technology-based businesses: What can emerging economies learn from the Western experience?' in Oakey, R., Groen, A., Cook, G. and Van Der Sijde, P. (eds.) *New technology-based firms in the new millennium* Emerald Group Publishing Limited, Leeds, pp. 183-196. doi:10.1108/S1876-0228(2013)0000010012.

Yao, A., Chan, N. and Yao, N. (2024) 'Understanding consumer behavior in phygital environments: An interpretivist methodological framework', *Qualitative Market*

Research: An International Journal, 27(3), pp. 449-470. doi:10.1108/QMR-08-2023-0100.

Appendix

Participant Information:

- Name:
- Country:
- Industry/Sector:
- Business Size (e.g., number of employees):
- Years in Operation:

A. Background Information

- 1. Briefly describe your business and the sector it operates in.
- 2. How long have you been operating your business?
- 3. What was the impact of the COVID-19 pandemic on your business (e.g., sales, operations, staffing)?

B. Impact of COVID-19 on Business Operations

- 1. How did the COVID-19 pandemic influence your approach to your business? (e.g., innovation, adaptation, risk management)
- 2. Did the pandemic create any new opportunities for your business? If yes, how did you adapt to capitalize on them?
- 3. In your opinion, what are the key qualities or skills that helped you navigate the challenges of the pandemic?

C. Government Policies and Support

- 1. Are you aware of any government policies or initiatives introduced during the pandemic to support businesses like yours? (If yes, please specify which ones)
- 2. Did you avail yourself of any of these policies or programs? If yes, which ones and how did they benefit your business?
- 3. Were there any challenges you faced in accessing or utilizing these programs?

D. Economic Recovery

- 1. What are your views on the economic recovery in your country post-pandemic?
- 2. How optimistic are you about the future of entrepreneurship in your country?

E. Policy Suggestions

- 1. What recommendations would you give to policymakers to better support entrepreneurs in times of crisis?
- 2. How can government policies be improved to foster a more resilient entrepreneurial ecosystem?

F. Additional Information:

Is there anything else you would like to add about your experience or the impact of government policies on your business?